Author’s response to reviews

Title: Comparison of Various Anthropometric Indices in predicting abdominal obesity in Chinese children: A cross-sectional study

Authors:

Gengdong Chen (chgengd@163.com)
Huanchang Yan (yhc117@foxmail.com)
Yuting Hao (hyt1025zy@163.com)
Shiksha Shrestha (sthшка_shikshya@hotmail.com)
Jue Wang (wangjue7952299@163.com)
Yan Li (pahliyan@163.com)
Yuanhuan Wei (wei20160915@163.com)
Zheqing Zhang (zzqaa501@smu.edu.cn)

Version: 1 Date: 20 Nov 2018

Author’s response to reviews:

Editor: 1. Thank you for providing information regarding consent from the participants. However, we would expect a parent or legal guardian to provide consent to participate for all participants in your study. Please clarify in your Ethics approval and consent to participate subheading whether the parents or legal guardians of all of the participants provided consent on their behalf.

Author: We added the information in the Ethics approval and consent to participate subheading, thank you.

2. Please add a "Declarations" heading between the list of Abbreviations and the "Acknowledgements" section.

Author: Added, thank you.
3. Please correct the numbering of the Tables. Currently, there are no tables labeled as Table 3 or Table 4.

Author: We have carefully checked the manuscript, and corrected the numbering of the Tables.

4. At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.

Author: We submitted a clean version, thank you.